Author's response to reviews

**Title:** Polymorphisms of interleukin-21 and interleukin-21-receptor genes confer risk for autoimmune thyroid diseases

**Authors:**

Zhang Jian (zhangjian890@126.com)
Xiao W Xia (xiaowanxia2005@163.com)
Zhu Y Feng (349276797@qq.com)
Muhali S Fatuma (muhalifatma@yahoo.co.uk)
Xiao Ling (xl77215@163.com)
Jiang W Juan (hidyfan81@yahoo.com.cn)
Shi X Hong (jjzhao@medmail.com.cn)
Zhou L Hua (zhoulianhua@medmail.com.cn)
Zhang J An (zhangjinan@hotmail.com)

**Version:** 2 **Date:** 22 February 2013

**Author's response to reviews:** see over
Polymorphisms of interleukin-21 and interleukin-21-receptor genes confer risk for autoimmune thyroid diseases

1. Department of Clinical Laboratory, Jinshan Hospital of Fudan University, Shanghai 201508, China
2. Internal Medicine Department, Xi’an Aviation Group Hospital, Xi’an, 710021, China
3. Endocrinology Department, Jinshan Hospital of Fudan University, Shanghai 201508, China
4. Endocrinology Department, Weinan Central Hospital, Weinan, Shaanxi, 714000, China

Correspondence: Jin-an ZHANG, Endocrinology Department, Jinshan Hospital, Fudan University, 1508 Longhang Road, Shanghai, P.R. China, Post code: 201508. Email: zhangjinan@hotmail.com
Tel: +086-021-34189990-5360

Abstract
Objective: The abnormality of interleukin-21 (IL-21)-IL-21-receptor (IL-21R) system has been found in many autoimmunity diseases including autoimmune thyroid diseases (AITD). In this study, we investigated whether polymorphisms of the IL-21 and IL-21R are associated with Graves' disease (GD) and Hashimoto's thyroiditis (HT), two major forms of AITD, among a Chinese population.

Design and Methods: Rs907715, rs4833837, rs2221903 and rs2055979 of the IL-21 gene and rs3093301 and rs2285452 of the IL-21R gene were explored in a case-control study including 405 GD, 228 HT patients and 242 controls. These genes were genotyped by PCR and restriction fragment length polymorphism (RFLP) analysis and the MASS spectrometry method.

Results: For IL-21 gene, we identified and confirmed a higher prevalence of A alleles of rs2221903 (P=0.018, OR=1.50 95%CI=1.07-2.09) in GD patients. We also found a significant association between rs2221903 and HT (allele: P=0.009; genotype: recessive P=0.021). For the IL-21R gene, compared with controls, the genotype frequencies of rs3093301 and rs2285452 were significantly different in HT patients using dominant genetic model (P=0.023, OR=1.61 95%CI=1.07-2.42; P=0.031, OR=1.71 95%CI=1.05-2.80, respectively). Furthermore, the haplotype AA containing the major alleles of rs4833837 and rs2221903 was associated with increased susceptibility to GD with an OR of 1.50(95%CI =1.08-2.09, P = 0.016), and to HT with an OR of 1.69(95%CI =1.14-2.52, P = 0.009).

Conclusion: Our results indicated that the SNPs of the IL-21 gene is associated with the development of GD. In addition, we found that individuals with the SNPs of the common IL-21 and IL-21R may have higher risk of HT.

Keywords: interleukin-21 (IL-21); interleukin-21 receptor (IL-21R); Graves’ disease; Hashimoto’s thyroiditis; single nucleotide polymorphism (SNP)

1. Introduction
Autoimmune thyroid diseases (AITD), the thyroid-specific autoimmune disorders, affect about 5% of the population. The etiology of AITD includes genetic and environmental factors resulting in immune abnormality and the development of AITD.
In 2000, the interleukin-21(IL-21)-IL21-receptor (IL-21R) system was discovered [1]. IL-21 and IL-21R are located on human chromosomes 4q26-27 and 16p11, respectively. The IL-21, preferentially produced by CD4+ T cells, has various functions, such as driving B cells to differentiate into memory cells and ultimately plasma cells, augmenting T cells’ proliferation, and promoting the activity of natural killer (NK) cells [2]. In accordance with these roles of IL-21, IL-21R is mainly expressed in B cells, T cells and NK cells, keratinocytes, and some of myeloid cells. Animal and clinical studies demonstrated the dysregulations of IL-21 and IL-21R in autoimmune diseases. For example, compared with IL-21R-competent BXSB-Yaa mice for multiple parameters of SLE, the IL-21R-deficient Yaa mice showed none of the abnormalities characteristic of systemic lupus erythematosus (SLE) [3]. IL-21R is overexpressed in the inflamed synovial membrane and in peripheral blood or synovial fluid leukocytes of rheumatoid arthritis (RA) patients [4]. And the evident increase of serum IL-21 level in primary Sjogren's syndrome (pSS) patients has positive correlation with the levels of gamma-globulin and erythrocyte sedimentation rate[5].These suggests that IL-21 and IL-21R may play a critical role in the pathogenesis of autoimmune diseases. Recently, H.Y. Jia [6] reported that an increase of serum IL-21 level in patients with primary GD.

In recent years, the associations between the IL-21 gene or IL-21R gene polymorphisms and autoimmune diseases have been gradually reported. In a HapMap-CEU population, a large block (480kb) of linkage disequilibrium in encompassing KIAA1109/Tenr/IL2/IL21 showed genetic associations with type 1 diabetes mellitus (T1DM) [7,8], RA [8], juvenile idiopathic arthritis (JIA) [9], psoriasis and psoriatic arthritis (PA) [10]. Similarly, the polymorphisms of the IL2/IL21 gene region likely to confer to the susceptibility to coeliac disease in Scandinavia families [11] and a US population [12], to ulcerative colitis in a North American and an Italian cohort [13], and to multiple sclerosis in a Spanish population [14]. After the SNPs in the IL-21 gene were analyzed, the significant differences of rs907715 and rs2221903 allele and genotype frequency were found between SLE and controls [15]. In addition to these reports, an association between microsatellite polymorphisms of the IL-21R and diabetes in Japanese patients between rs3093301 and rs2285452 in the IL-21R gene region and SLE in the European-derived cohort were documented [16].

Therefore, it is reasonable to speculate that IL-21 and IL-21R are the candidate genes for AITD. Plagnol [17] reported that the rs2069763 in chromosome 4q27 were associated with the GD susceptibility in a Caucasian cohort. And a recent study that indicated that rs907715 of IL-21 gene was associated with GD in a Chinese population [6].

In this study, additional variants of the IL-21 gene and two SNPs of the IL-21R gene in addition to previously reported variants were investigated in two subtypes of AITD including GD and HT.

2. Materials and methods
2.1 Patients and controls

AITD patients and healthy controls were enrolled from the First Affiliated Hospital, Medical School of Xi’an Jiaotong University. A total of 633 independent patients with AITD, including 405 GD and 228 HT, and 242 unrelated healthy controls were recruited. GD and HT were diagnosed based on clinical and laboratory evidence of hyperthyroidism and hypothyroidism respectively and diffuse goitre, supported by the presence of antithyroglobulin antibody (TgAb) and/or antithyroid peroxidase antibody (TPOAb) and/or exophthalmos. The controls were healthy subjects without clinical evidence or family history of any AITD. This study was approved by the ethics committee of Jinshan hospital of Fudan University. All patients and control subjects were asked to sign an informed consent.

2.2 Genotyping

Genomic DNA was extracted from peripheral blood leukocytes using the Nucleon Bacc kit from TianGen Biotech CO. LTD (Beijing, China).

Rs907715, rs4833837, rs2221903 and rs2055979 spanning a large region which captured all common variation in the IL-21 gene and rs3093301 and rs2285452 of IL-21R were selected from the published SNP database (http://www.ncbi.nlm.nih.gov/project/SNP). All of these SNPs were validated polymorphisms, and have a minor allele frequency of more than 1%. Rs907715, rs4833837, and rs222190 are located on intron 2 of IL-21R and rs2055979 on intron 3. For IL-21R gene, rs3093301 is in intron 2 and rs2285452 in exon 10.

All of the four IL-21 SNPs and rs2285452 were typed using mass spectrometry method (Shanghai Benegene Biotechnologies CO. Ltd. Shanghai, China). Rs3093301 was typed using PCR-restriction fragment length polymorphism (PCR-RELP) method. The appropriate fragment of rs3093301 in the IL-21R gene was amplified using specific primers (forward: 5’-AATTGCTCCT CAGCAGA TCC-3’, reverse: 5’-GCA TCAGCCTC CCGAGTAG-3’). The PCR reaction was performed in a 25µl mixture, containing 100ng genomic DNA and 0.1 mM solution of each primer, 12.5µl 2*Taq PCR MasterMix (TianGen Biotech CO. LTD, Beijing, China) and 9.5µl ddH2O. PCR was carried out with initial denaturation for 5 min at 95°C, followed by 37 cycles of annealing for 30 seconds at 60°C, extension for 30 seconds at 72°C and denaturation for 30 seconds, and a final extension for 5 min at 72°C. The product (383bp) was digested with restriction enzyme NlaIII (Fermentas Life Sciences) at 37 centigrade for 10 hours, and analyzed on a 3% agarose gel. There were three profiles of the digested fragments: 210bp, 44bp, and129bp indicating the presence of TT, two fragments of 254bp and 129bp indicating the presence of CC, and four fragments of 254bp, 210bp, 129bp and 44bp representing TC genotype.

2.3 Statistical analysis

In this study, we used a population-based case-control method. Statistical analysis was performed using Haploview 4.1 and SPSS 11.0. Hardy-Weinberg equilibrium (HWE) of each SNP
was analyzed in controls using Goodness-of-fit test. Categorical data were analyzed by chi-square or Fisher’s exact test. For continuous data, Student’s t-test was used for the normalized data, and non-parametric test was used for non-normalized data. Furthermore, corrected P value and gene-gene interaction were analyzed using logistic regression analysis. In this study, a two-tailed P value less than 0.05 was considered to be significantly difference.

3. Results
Four SNPs (rs907715, rs4833837, rs2221903 and rs2055979) of the IL-21 gene and two SNPs (rs3093301 and rs2285452) of the IL-21R gene were typed in 405 GD and 228 HT patients and 242 controls. For the tested SNPs, genotyping successful rate was more than 96.9%. All of the 6 genotyped SNPs showed a minor allele frequency more than 7% and HWE analysis in controls showed P values of more than 0.05. When comparing age and sex distributions between patients and controls, we found the control group matched well with the GD group, but not with HT group (seen in Table 1). The distributions of the SNPs genotypes in HT patients were analyzed using logistic regression analysis adjusted for sex and age when compared with the controls.

3.1 Association of the IL-21 gene polymorphisms with GD and HT
A genetic association between GD and HT and one of the four genotyped IL-21 SNPs was found in this study. For the SNP rs2221903 located in the intron 2 regions, the frequency of A allele was 89.2% in GD and 90.4% in HT, respectively, significantly higher than 84.7% in healthy controls (GD: P=0.018, OR=1.495 95%CI=1.070-2.086; HT: P=0.009, OR=1.69 95%CI=1.13-2.51). In contrast to these results, the allele frequencies of rs907715, rs4833837, rs2221903 and rs2055979 were similar between patients and controls (Table 2). As the distributions of rs907715, rs4833837, rs2221903 and rs2055979 genotypes in patients and controls are shown in Table 3. For rs2221903, the frequencies of the genotypes carrying risk allele (A) were higher both in GD (AA: 80.2%, GA: 18.1%) and HT (AA: 81.1%, GA:18.4%) patients than in controls (AA:73.6%, GA:22.2%), but statistical significance was only found when the recessive genetic model was used in HT patients (AA+GA vs. GG, P=0.021, OR=11.72 95%CI=1.46-94.13), in comparison with the controls. In this study, no significant association was observed between GD or HT and rs907715, rs4833837 or rs2055979 genotypes while different genetic models were applied for analysis.

3.2 Association of the IL-21R gene polymorphisms with GD and HT
As shown in Table 2, the rs3093301 and rs2285452 SNPs in IL-21R showed no association with GD or HT (P=0.797-0.068).

Nominal associations were found between rs3093301 and rs2285452 and HT when the dominant genetic model (CC vs. TC+TT, P = 0.023, OR=1.61 95%CI= 1.070-2.42); GG vs. GT+TT, P = 0.031, OR=1.71 95%CI=1.05-2.80, respectively) was used for analysis. Genotype analysis of rs3093301 and rs2285452 using additive and recessive genetic model did not indicate
any significant difference between HT patients and controls (Table 3). For rs3093301 and rs2285452, no significant different distribution was found in GD patients and controls when different genetic models was used (showed by Table 3).

3.3 Haplotype analysis in GD and HT
As shown in Table 4, rs4833837 and rs2221903 (r-square = 0.901) were identified as an LD block using the Haploview 4.1. The total frequency of haplotypes listed in Table 5 was more than 99.0% in every group. AA containing major alleles of these two SNPs was associated with increased susceptibility to GD with an OR of 1.50 (95% CI = 1.08-2.09, P = 0.016), and to HT with an OR of 1.69 (95% CI = 1.14-2.52, P = 0.009) (Table 5).

4. Discussion
IL-21 is a member of the common-gamma chain family of cytokines with immunoregulatory activity. IL-21R has been shown to form a heterodimeric receptor complex with the common gamma-chain. IL-21 binding with this receptor leads to the activation of multiple downstream signaling molecules, including JAK1, JAK3, STAT1 and STAT3, therefore affects the innate and adaptive immune responses by inducing the differentiation, proliferation and activity of multiple target cells. The dysregulation of IL-21 and IL-21R plays a role in multiple immune-mediated diseases, including SLE [3], psoriasis [5], RA [4] and other chronic inflammatory diseases [18]. Like other autoimmune diseases, GD and HT are chronic diseases initiated by the loss of immunological tolerance to self-antigens. Previous studies indicate that some immune-related genes may participate in the development of AITD.

In this study, we found a significant association between GD or HT and rs2221903 located in the IL-21 gene region, and the frequencies of the haplotypes in the IL-21 gene that consists of rs4833837 and rs2221903 in GD and HT patients were significantly different from controls. For rs3093301 and rs2285452 polymorphisms of the IL-21R gene, there were significantly different distributions of these genotypes between HT patients and the controls.

In different from Jia’s study [6], we did not find the rs907715 SNP linkage with GD. In Jia’s report, the AA genotype frequency was lower in GD patients (19.4%) than in controls (30%). The gene frequency in patients of our study was similar to that of Jia’s study, but different in the two control groups. In our study the frequency of AA genotype in the control group was 18.9%, which is similar to the frequency in the HapMap-CHB population. This may be the geographical or selection variation that results in this difference. It should be noted that the lack of association in our study does not completely exclude the possibility of IL-21R as a candidate gene for GD because of the following two reasons: 1) The minor allele frequency is 0.2, which gives our study genetic power of about 0.8 with an OR of homozygote 2.0, and of heterozygote 1.5, therefore studies with a larger sample size are necessary to confirm whether patients with these SNPs have more risk for development of GD. 2) As many other immunologic disorders, AITD is
believed to derive from a multiple network of various susceptible loci, which exert synergic or additive effects, but each locus may play a small role [19].

In genetics, a recessive gene is an allele that causes a phenotype (visible or detectable characteristic) that is only seen in a homozygous genotype (an organism that has two copies of the same allele) and never in a heterozygous genotype. While dominance is a relationship between alleles of a gene, in which one allele masks the expression (phenotype) of another allele at the same locus. Our study didn't find any synergistic effect of the risk genotypes and we were unable to conclude whether these two genes were either recessive or dominant over the other gene.

IL-21 is located on chromosome 4q26-27, which is close to IL-2, a region that has been linked to AITD susceptibility [17]. Jia’s et al found rs907715 of IL-21 gene being associated with GD in a Chinese population while Plagnol found only rs2069763 and did not find association of rs2069762 and rs6822844 of IL-2 with GD, maybe there is ethnicity or regional effect contributing to these results differences. Therefore, we can not exclude that the SNPs of the IL-21 and IL-21R genes are only an indicator of a candidate gene contributing to AITD. Further studies such as identifying the causal SNP for this association, gene-environment interaction in GD and HT cohort studies required to clarify the effect of IL-21 and IL-21R on AITD susceptibility. Replication of these associations between IL-21 and IL-21R gene and AITD is also required in larger independent databases of different cohorts.

In conclusion, our study confirmed the synergic effect of the IL-21 SNPs in the development of GD. In addition, to the best of our knowledge, we are the first to report the association of the IL-21 and IL-21R SNPs with an increased risk of HT.

Declaration of Interest

The authors report no conflict of interest. The authors alone are responsible for the content and writing of the paper.

Authors’ contributions

WXX and YFZ recruited the subjects and participated in the sequence alignment. LX and WJJ extracted genomic DNA from peripheral blood leukocytes. XHS carried out the molecular genetic studies, participated in the sequence alignment. JZ carried out the immunoassays and drafted the manuscript. SFM participated in the design of the study and performed the statistical analysis. JAZ and LHZ conceived of the study, and participated in its design and coordination. All authors read and approved the final manuscript.

Acknowledgments

This work was supported by grants from the National Natural Science Foundation of China (30871184, 81070627).
References


